Author's response to reviews

Title: Incidence and prevalence of upper-extremity musculoskeletal disorders. A systematic appraisal of the literature.

Authors:

Bionka MA Huisstede (b.huisstede@erasmusmc.nl)
Sita MA Bierma-Zeinstra (s.bierma-zeinstra@erasmusmc.nl)
Bart W Koes (b.koes@erasmusmc.nl)
Jan AN Verhaar (j.verhaar@erasmusmc.nl)

Version: 5 Date: 16 January 2006

Author's response to reviews:

1. I removed the keywords section from the manuscript
2. I oriented the tables in portrait rather than in landscape